Author's response to reviews

Title: The effectiveness of a semi-tailored facilitator-based intervention to optimise chronic care management in general practice: a stepped-wedge randomised controlled trial

Authors:

Tina Drud TDD Due (tina.due@sund.ku.dk)
Thorkil TT Thorsen (tthorsen@sund.ku.dk)
Marius Brostrøm MBK Kousgaard (marbro@sund.ku.dk)
Volkert Dirk VDS Siersma (siersma@sund.ku.dk)
Frans Boch FBW Waldorff (fransw@sund.ku.dk)

Version: 4 Date: 27 March 2014

Author's response to reviews: see over
Dear Magdalena Morawska

Thank you for the opportunity to revise our manuscript entitled “The effectiveness of a semi-tailored facilitator-based intervention to optimise chronic care management in general practice: a stepped-wedge randomised controlled trial” (MS: 1729622196117865).

We much appreciate the constructive comments of the reviewers and we have revised the manuscript accordingly, paying particular attention to the four points highlighted by the editors. The manuscript has also been edited by a native English-speaking professional language editor.

We have uploaded six documents:
1. A point-by-point response to the comments of the reviewers.
2. A revised version of the paper.
3. An updated CONSORT-checklist.
4. An revised trial flow (the number of practices not included in the study was not visible in the former version)
5. An revised table 1 and table 2 (the unit in the primary outcome has been changes from number of check-ups per patients in the practice to number per 100 patients in the practice)
6. An revised table 2 (the same change as in table 1)

Yours sincerely
Tina Drud Due